Author's response to reviews

Title: Prevalence of maternal smoking and environmental tobacco smoke exposure during pregnancy and impact on birth weight: retrospective study using Millennium Cohort

Authors:

Corinne Ward (mzyxcmw@nottingham.ac.uk)
Sarah Lewis (Sarah.lewis@nottingham.ac.uk)
Tim Coleman (Tim.coleman@nottingham.ac.uk)

Version: 4 Date: 16 May 2007

Author's response to reviews:

All requested changes have been made.